

Geophysical Research Letters^{*}

RESEARCH LETTER

10.1029/2021GL095011

Key Points:

- Atmospheric circulation anomalies associated with the most severe haze pollution are identified
- Haze formation in eastern China has shifted from causes of local accumulation in northern China to regional transport in southern China
- Climate change under the sustainable and intermediate development scenarios are the ideal paths to reduce haze in China

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

Y. Yang, yang.yang@nuist.edu.cn

Citation:

Yang, Y., Zhou, Y., Li, K., Wang, H., Ren, L., Zeng, L., et al. (2021). Atmospheric circulation patterns conducive to severe haze in eastern China have shifted under climate change. *Geophysical Research Letters*, 48, e2021GL095011. https://doi. org/10.1029/2021GL095011

Received 25 JUN 2021 Accepted 22 NOV 2021

© 2021. American Geophysical Union. All Rights Reserved.

Atmospheric Circulation Patterns Conducive to Severe Haze in Eastern China Have Shifted Under Climate Change

Yang Yang¹, Yang Zhou¹, Ke Li², Hailong Wang³, Lili Ren¹, Liangying Zeng¹, Huimin Li¹, Pinya Wang¹, Baojie Li¹, and Hong Liao¹

¹Jiangsu Key Laboratory of Atmospheric Environment Monitoring and Pollution Control, Jiangsu Collaborative Innovation Center of Atmospheric Environment and Equipment Technology, School of Environmental Science and Engineering, Nanjing University of Information Science and Technology, Nanjing, China, ²Harvard John A. Paulson School of Engineering and Applied Sciences, Harvard University, Cambridge, MA, USA, ³Atmospheric Sciences and Global Change Division, Pacific Northwest National Laboratory, Richland, WA, USA

Abstract Regional atmospheric circulation patterns affect haze pollution and they change in the warming climate. Here, the characteristics of atmospheric circulation anomalies conducive to extreme haze occurrence in China and their historical and future trends are examined based on surface observations, reanalysis data, aerosol source tagging technique, and multimodel intercomparison results. December 2016 and 2017 are identified as the worst months of haze pollution over northern and southern China, featuring weakened and strengthened prevailing winds, respectively. During 1980–2019, the atmospheric pattern similar to December 2016 decreased, while that similar to 2017 increased, suggesting that severe haze formation mechanism in eastern China has been shifting from causes of local accumulation to regional transport processes. In the future, climate change under the sustainable and intermediate development scenarios are the ideal paths to reduce haze in China, while high social vulnerability and radiative forcing would cause a severe damage to the environment.

Plain Language Summary Severe haze pollution occurred in December 2016 and 2017 over different regions of China related to local accumulation and regional transport of aerosols, respectively. During the past four decades, the frequency of atmospheric circulation patterns similar to December 2016 decreased and that similar to 2017 increased. It indicates that haze formation in eastern China has shifted from local accumulation in northern China to regional transport in southern China. The ideal path for China to reduce haze is under sustainable development scenarios whereas high levels of social vulnerability and radiative forcing will cause severe damage to the atmospheric environment in the future.

1. Introduction

With the rapid industrialization and urbanization, haze pollution has become more frequent and serious in China during the recent decades (Reddington et al., 2019). When severe haze hit China in the past, the maximum daily $PM_{2.5}$ (particulate matter less than 2.5 µm in diameter) concentration reached 500 µg m⁻³ in eastern China, which was 20 times higher than the healthy air quality criterion of World Health Organization (Li et al., 2018). $PM_{2.5}$ has various effects on environment and climate. $PM_{2.5}$ harms human health causing cardiovascular and respiratory disease and shortens life expectancy (Zhang et al., 2017). In China, about 1 million people died every year due to $PM_{2.5}$ exposure (Burnett et al., 2018; Cohen et al., 2017). On the other hand, through aerosol-radiation and aerosol-cloud interactions, $PM_{2.5}$ can directly and indirectly affect climate (Yang, Ren, et al., 2020; Yang et al., 2019). Moreover, high concentration of $PM_{2.5}$ in hazing conditions reduces atmospheric visibility and thus endangers road traffic and air transportation (Ding & Liu, 2014; Zhang et al., 2014). Therefore, it is imminent to study the causes of haze in China and the past and future changes of these factors.

Numerous studies have identified anthropogenic emissions (Fu et al., 2016; Reddington et al., 2019; Sun et al., 2018; Yang et al., 2016) and adverse meteorological conditions (Cai et al., 2017; Ding & Liu, 2014; Pei et al., 2020; Wu et al., 2017; Zou et al., 2017) as two key factors leading to the frequent occurrence of severe wintertime haze events in China. Since the promulgation of Air Pollution Prevention and Control Action Plan, the emissions of pollutants in China have significantly decreased (Li et al., 2021; Zheng et al., 2018) and the impact of meteorological conditions on haze pollution has been attracting more attention. Adverse meteorological conditions, including weak winds, low boundary layer height, strong temperature inversions, and high relative

humidity closely associated with anticyclonic anomalies over northeastern Asia, can lead to severe haze pollution in northern China (Zhang et al., 2016; Zhong et al., 2019). After the cold frontal transport from the North China Plain to the Yangtze River Delta, $PM_{2.5}$ can accumulate under the control of local high pressure to worsen the air quality in the Yangtze River Delta region (Kang et al., 2019). Li et al. (2018) reported that haze in eastern China was often related to an anomalous high sea level pressure in northeastern Asia and the Pacific Northwest and anomalous high pressure that controlled the East Asian trough or the whole East Asia in the mid-troposphere, resulting in weakened near-surface northwest prevailing winds during winter.

These previous studies provided important information that atmospheric pattern anomalies have largely influenced haze formation in China; however, most studies only paid close attention to a certain region and lacked comparative analyses on the atmospheric patterns and relevant haze events in various regions of China. In addition, aerosol and their precursor emissions are changing rapidly in recent years in part due to clean air actions. The compound effects of emission changes should be screened out in order to understand the impacts of atmospheric conditions on haze pollution. Furthermore, using observational data alone, it is difficult to separately quantify contributions from local emissions/accumulation and regional transport driven by certain atmospheric patterns. Moreover, past and future changes of these atmospheric patterns conducive to severe haze under climate change, which are critical to pollution prevention, deserve more investigation.

Here, PM_{2.5} observations, reanalysis data and multimodel intercomparison results and an aerosol-climate model with the Explicit Aerosol Source Tagging (EAST) method are used to explore the relationship between the atmospheric circulation patterns and heavy pollution events in China, which is of great significance for the control of air pollution, climate change mitigation, and the sustainable development of economy and society.

2. Materials and Methods

In this study, observed hourly $PM_{2.5}$ concentrations in December–January–February (DJF) for years 2009–2019 obtained by the U.S. Embassy in Beijing, Shanghai, and Guangzhou cities (Figure S1 in Supporting Information S1), respectively, located in the three subregions of China (North China Plain [NCP] and the Yangtze River Delta [YRD]/Pearl River Delta [PRD]) are used to characterize haze events. The rapid industrial and economic development and unfavorable meteorological conditions in these three regions had been reported to lead to frequent haze events (Li et al., 2016; Liao et al., 2017; Liu et al., 2019; Mai et al., 2016), and hence it is necessary to examine the atmospheric circulation patterns in these three typical polluted regions to provide a basis for the overall planning of future pollution prevention and control measures in each region. To remove the impact of emission changes on $PM_{2.5}$ variations in observational analysis, the observed $PM_{2.5}$ concentrations are normalized by the monthly SO₂ emissions (precursor gas of sulphate aerosol) over the three polluted regions since that sulphate is a large contributor of $PM_{2.5}$ mass (Cao et al., 2012) and the driver of $PM_{2.5}$ variation in China (Geng et al., 2017). The SO₂ emissions are from the Multi-resolution Emission Inventory for China (MEIC) inventory over 2009–2017 with 2018–2019 data obtained by a linear interpolation. Severe haze day is defined by daily $PM_{2.5}$ concentration greater than 150 µg m⁻³.

A global aerosol–climate model CAM5 with the Explicit Aerosol Source Tagging (CAM5-EAST, Wang et al., 2014; Yang, Wang, Smith, Ma, & Rasch, 2017, Yang, Wang, Smith, Easter, et al., 2017) is applied in this study to quantitatively attribute the sources of aerosols from different regions. Aerosols from different source regions of interest are explicitly tracked in the CAM5-EAST simulation without perturbing the emissions. The contributions of many source regions including subregions of China and rest of the world to the PM_{2.5} concentrations are quantified to explore the source-receptor relationship in the three cities and are shown in Figure S1 in Supporting Information S1. The CAM5-EAST simulation is performed to cover the focused historical period. Detailed information about the EAST technique, implementation in CAM5 and model evaluation can be found in the previous studies (Ren et al., 2020; Wang et al., 2014; Yang, Lou, et al., 2020; Yang, Wang, Smith, Zhang, Lou, Qian, et al., 2018; Yang, Wang, Smith, Zhang, Lou, Yu, et al., 2018).

To calculate the anomalies of meteorological fields during severe haze pollution events, ERA5 reanalysis data are adopted from 1980 to 2019. Following Li et al. (2018), monthly meteorological variables including sea level pressure (SLP), wind fields (at 850 and 500 hPa) and geopotential height (GPH) at 500 hPa are selected to determine anomalous atmospheric circulation patterns and their variations under climate change.





Figure 1. Time series of frequency of severe haze (days in each month with daily $PM_{2.5}$ concentration greater than 150 µg m⁻³) in Beijing, Shanghai and Guangzhou (a–c), after the influence of emission changes is removed by normalizing the frequency to monthly SO₂ emissions. The dark-colored bars represent the most severe months having the highest frequencies of hazy days.

For the trends in circulation patterns conducive to severe haze under future climate change, the multimodel future simulations from the Scenario Model Intercomparison Project (ScenarioMIP) under the Coupled Model Intercomparison Project Phase 6 (CMIP6) are utilized, together with the CMIP6 historical results as baseline cases. Four future scenarios for the Shared Socioeconomic Paths (SSPs), including SSP1-2.6, SSP2-4.5, SSP3-7.0, and SSP5-8.5 are the focus of this study, allowing for comprehensive analysis of future changes in meteorological conditions and air pollutants. Totally 25 models, including ACCESS-CM2, ACCESS-ESM1-5, AWI-CM-1-1-MR, BCC-CSM2-MR, CAMS-CSM1-0, CESM2-WACCM, CMCC-CM2-SR5, CanESM5, EC-Earth3, EC-Earth3-Veg, EC-Earth3-Veg-LR, FGOALS-f3-L, FGOALS-g3, GFDL-ESM4, IITM-ESM, INM-CM4-8, INM-CM5-0, IPSL-CM6A-LR, MIROC6, MPI-ESM1-2-HR, MPI-ESM1-2-LR, MRI-ESM2-0, NorESM2-LM, NorESM2-MM, TaiESM1, are used to analyze future trends of anomalous circulation patterns.

3. Results

3.1. Atmospheric Circulation Patterns Conducive to Severe Haze

The frequency of severe winter haze days during 2009–2019 in Beijing, Shanghai, and Guangzhou representing the top polluted three subregions of China is shown in Figure 1. The most frequent haze pollution in Beijing occurred in December 2016, with 18 haze days in the month. The most frequent haze pollution in Shanghai and Guangzhou during the observed time period was in December 2017, when Shanghai and Guangzhou had 9 and 4 haze days, respectively. The frequency was less than that in Beijing but not neglected given that the background level of $PM_{2.5}$ is normally low. Although the emissions of air pollutants were significantly reduced, the severe haze in winter has not been much improved, indicating that large-scale circulation anomalies played an important role (Krotkov et al., 2016). In this study, December 2016 and 2017 are used as the most severe haze months to analyze for the relationship between atmospheric circulation anomalies and severe haze pollution over NCP in northern China and YRD/PRD in southern China, respectively, under climate change.

Atmospheric circulation anomalies that are conducive to the extreme haze over China in December 2016 and 2017 are characterized in Figure 2. When the most severe haze occurred in NCP (December 2016), an abnormal SLP gradient between the low pressure over China and the high extended from Northwest Pacific to northeastern China, relative to the 40-year climotology during 1980–2019, was observed (Figure 2a). The abnormal pressure gradient reduced the prevailing northwesterly winds in the lower troposphere in winter over the NCP region. In the mid-troposphere, an abnormal anticyclone appeared over NCP (Figure 2b). Under the control of this anomalous high pressure, the atmospheric boundary layer was stable and the prevailing northwesterlies were pushed to the north, leading to poor dispersion conditions, local aerosol accumulation, and haze formation in NCP.

The atmospheric circulation anomaly in December 2017 was significantly different from that in December 2016. In December 2017, a low pressure located over the Sea of Japan and northeastern China along with an anomalous high pressure over China in both low- and mid-troposphere, which strengthened the prevailing northerly winds in eastern China (Figures 2c and 2d). As a result of the northerly wind anomalies, more aerosols were transported



Figure 2. Atmospheric circulation anomalies observed in December 2016 and December 2017 (a)–(d). Observed anomalies in sea level pressure (SLP, hPa, shaded) and 850 hPa winds (m s⁻¹, vector) (a), and geopotential height (GPH, m, shaded) and winds at 500 hpa (m s⁻¹, vector) (b), in December 2016 relative to the 40-year average (1980–2019). (c) and (d) are same as (a) and (b), respectively, but for December 2017. The green boxes mark the cities, Beijing in (a) and (b) and Shanghai and Guangzhou in (c) and (d), that had haze pollution during the corresponding month.

from the polluted northern China to the south, accounting for the severe haze pollution over YRD and PRD in December 2017.

The sudden increases in aerosol concentrations to form severe haze pollution are either from local emissions due to a stagnant condition or from imported contribution by atmospheric transport related to the anomalous circulation patterns. To further examine the mechanism for the impact of atmospheric circulation anomaly on haze pollution, the CAM5-EAST model is used to quantify sources of aerosols in the three cities we focused on. Figure S2 in Supporting Information S1 shows the contributions of individual source regions to the differences in PM_{2.5} concentrations between the severe haze months and normal conditions, and the values are also summarized in Table S1 in Supporting Information S1.

The contribution of NCP local emissions accounts for 91% of the $PM_{2.5}$ concentration increase in Beijing during the most severe haze pollution in December 2016, consistent with the weakening of prevailing winds in NCP (Figure 2a). It indicates that local emissions are the dominant factor of haze pollution in NCP. Over Shanghai, NCP is the only positive contributor and emissions from NCP and southern China contribute 68% and 46% to the $PM_{2.5}$ increase in Guangzhou, respectively, in December 2017. It affirms that the most severe haze pollutions over YRD/PRD in southern China in December 2017 were due to the anomalously strong regional transport associated with the strengthened northerly winds.

3.2. Shifting Patterns From Local Accumulation to Regional Transport

Historical variations of the atmospheric circulation patterns conducive to severe haze pollution are essential for understanding the observed haze events during the past few decades in China. Figure 3 shows the correlation between SLP or 500 hPa GPH anomalies in December 2016/2017 and the same fields in each year during 1980–2019, as well as the frequency of the patterns of moderate to high correlation (>0.5) with those in December 2016/2017. It is interesting that both the frequencies of SLP and GPH that have moderate to high correlation with





Figure 3. Time series of pattern correlation between SLP (left) or 500 hPa GPH (middle) anomalies over East Asia and western Pacific $(90^{\circ}-160^{\circ}\text{E}, 20^{\circ}-60^{\circ}\text{N})$ in December 2016 (top)/2017 (bottom) and those in December of each year during 1980–2019. The dotted lines mark the correlation threshold of +0.5, which is used to define "moderate to high correlation." The bar chart (right) represents the frequency of SLP and 500 hPa GPH that have moderate to high correlation (>0.5) in December 2016/2017 during 1980–1999 (blue) and 2000–2019 (orange). Anomalies relative to the averages over 1980–2019 are used in the calculation of correlations.

those in December 2016 decreased during 1980–2019 (i.e., less frequent in 2000–2019 than in 1980–1999). It indicates that although the anthropogenic emissions have increased during the past four decades, the occurrence of circulation patterns conducive to extreme haze over NCP in northern China was decreasing, in favor of the regional air quality improvement. On the contrary, the frequency of atmospheric circulation anomalies similar to that in December 2017 increased significantly from 1980–1999 to 2000–2019. It implies that the variation in atmospheric circulation might have been responsible for the intensification of severe haze over YRD and PRD in southern China since 1980s.

The regional atmospheric circulation is affected by both natural climate internal changes and human factors, including Arctic Oscillation (Lu et al., 2020), Pacific Decadal Oscillation (Zhao et al., 2016), Arctic sea ice (Wang et al., 2015), and global warming (Cai et al., 2017). In the year with high Arctic Oscillation Index (AOI), the weakened East Asian trough causes a decrease of the meridional winds, making the meteorological condition unfavorable for the dispersion of air pollutants, which significantly impacts haze pollution in northern and central China (Lu et al., 2020). In 2016, AOI was greater than 1, explaining the anomalous atmospheric circulation pattern conducive to serious haze pollution in December 2016. In addition, the historical years with abnormally high AOI (>1) identified by Lu et al. (2020) were 1988, 1989, 1991, 1992, 1999, 2006, and 2016, which was likely the reason for the decrease in occurrence frequency of the circulation anomaly pattern in 1980–2019 that is similar to December 2016. In contrast, the December 2017 pattern with strengthened northerly winds increased, potentially due to the less frequent high-AOI years after 2000 compared to pre-2000.

Nevertheless, the frequency of atmospheric circulation pattern similar to December 2016 with weakened lower tropospheric winds is decreasing, while the December 2017 pattern with strengthened winds is increasing. The transition of atmospheric circulation anomalies from the 2016-type to the 2017-type suggests that the haze formation mechanism in eastern China is shifting from the local accumulation in northern China to regional transport in southern China under historical climate change.





December 2016

December 2017

Figure 4. Frequency of SLP and 500 hPa GPH that have moderate to high correlation (>0.5) in December 2016 (a–d)/2017 (e–h) in each 10-year interval during 2021–2100 under four SSPs furture scenarios of CMIP6. It is calculated by the sum of the moderate to high correlation coefficients of the 25 models for each decade. The four SSPs are SSP1-2.6, SSP2-4.5, SSP3-7.0, and SSP5-8.5 (from top to bottom). The slope and P values of linear regressions during 2021–2100 are shown in the upper right of each panel. The linear trends of SLP and GPH in each model grid were removed before the correlation coefficient is calculated.

3.3. Future Trends of Atmospheric Circulation Patterns

Projection of the trends of the atmospheric circulation patterns conducive to haze in China under future climate change is important for the policy making. Here, meteorological variables simulated by 25 climate models under four future climate policy scenarios, including SSP1-2.6, SSP2-4.5, SSP3-7.0, and SSP5-8.5 that represents sustainable, medium, local, and conventional development, respectively, are used to characterize the future regional circulation patterns. The frequencies of haze-conducive atmospheric patterns are shown in Figure 4. It should be noted that most of future trends are not statistically significant due to the diverse results of the 25 models analyzed in this study, so they only represent a possible future change for the consideration.

Under the SSP1-2.6 and SSP2-4.5 scenarios, both the frequencies of SLP and GPH patterns similar to those in December 2016 show downward trends during 2021–2100 (Figures 4a and 4b), whereas the changes are opposite

for SLP and GPH patterns under the SSP3-7.0 and SSP5-8.5 scenarios (Figures 4c and 4d). For the circulation patterns similar to that in December 2017 (Figures 4e–4h), only the SSP2-4.5 presents downward trends for both SLP and GPH over 2021–2100, while their trends are in the opposite directions under other scenarios. It suggests that the sustainable and medium development scenarios are the ideal paths to the fresh air in China from the perspective of circulation changes. The efforts for moderate to low social vulnerability and radiative forcing, international cooperation, land use change for more forests under SSP1-2.6 and SSP2-4.5 are essential for both the climate change mitigation and air quality improvement.

Figure S3 in Supporting Information S1 depicts the frequencies with correlation coefficients greater than 0.5 between the atmospheric circulation anomalies in December 2016/2017 and those in the historical 50 years (1965– 2014) and future 50 years (2045–2094) from the 25 CMIP6 climate models. Under the SSP3-7.0 and SSP5-8.5 scenarios, atmospheric circulation anomalies similar to those in December 2016 or 2017 are more frequent in the future than in the historical period, which can be partly attributed to the upward trends in the frequency and magnitude of Arctic Oscillation in a warmer change related to increases in anthropogenic greenhouse gases (Shindell et al., 1999). It indicates that vicious competition between nations, high social vulnerability, and high anthropogenic radiative forcing can lead not only to a slow economic growth and warmer climate, but also to a severe environmental damage. The importance of international win-win cooperation and the development of green economy should always be promoted.

4. Conclusions and Discussions

High concentration of PM_{2.5} is extremely harmful to human health, atmospheric visibility, green economy, and other environmental/societal aspects. This paper studies the characteristics of atmospheric circulation anomalies during the most severe wintertime haze pollution and historical and future trends of relevant atmospheric circulation patterns that are conducive to the haze pollution in China under the changing climate. After the impact of emission changes being removed, December 2016 and 2017 are identified to be the most severe haze months over NCP and YRD/PRD, respectively. The atmospheric circulation anomaly in December 2016 was characterized by an anomalous SLP gradient between a low pressure over China and a high pressure extended from Northwest Pacific to northeastern China, which reduces the lower tropospheric northwesterly winds over the NCP region, together with the stable boundary-layer condition under high pressure, leading to poor dispersion conditions. In contrast, the December 2017 atmospheric circulation anomaly is characterized by a low pressure over the Sea of Japan and northeastern China, along with a high pressure over China in both the low- and mid-troposphere, strengthening the prevailing northerly winds in eastern and southern China. The northerly wind anomaly leads to aerosol transport from polluted northern China to the south.

After studying the historical variations of the atmospheric circulation patterns leading to extreme haze pollution, we find that the atmospheric circulation pattern with weakened winds similar to December 2016 has been decreasing during 1980–2019, while the pattern similar to December 2017 with strengthened winds has been increasing. The change in atmospheric circulation anomalies from the 2016-pattern to the 2017-pattern suggests that severe haze formation in eastern China is shifting from local accumulation in northern China to regional transport in southern China under the historical climate change.

Trends of atmospheric circulation patterns favoring haze in China under future climate change are also studied. Decreasing trends in frequencies of SLP and GPH patterns similar to those in December 2016/2017 are projected to occur under SSP1-2.6 or SSP2-4.5 scenarios. This suggests that sustainable and intermediate development scenarios are the preferred paths for both the global climate change mitigation and the clean air target in China in the future. Vicious competition between countries, high social vulnerability, and high anthropogenic radiative forcing (SSP3-7.0 and SSP5-8.5) in the future will not only lead to a slow economic growth and warmer climate, but also cause a severe damage to the environment compared to current days.

In this study, we found that the occurrence frequency of atmospheric patterns conducive to extreme haze in China is expected to increase under high emission-based climate change in the 21st century (SSP3-7.0 and SSP5-8.5), which is in accordance with Cai et al. (2017) that future weather conditions in favor of haze formation in Beijing would worsen under RCP8.5 (Representative Concentration Pathway 8.5) scenario. The increase was attributed to mean state changes, including accelerated temperature increases in the lower atmosphere, weakened East Asia winter monsoon, and increasing sea level pressures over the North Pacific related to the upward trends in

the frequency and magnitude of Arctic Oscillation. The decreasing trends under SSP1-2.6 or SSP2-4.5 may be influenced by effective environmental pollution control such as lower anthropogenic radiative forcing, lower CO_2 emissions, and a modest temperature increase. However, due to the differences in simulation performance between models, the future prediction needs further data and analysis to provide a reference for haze prevention and control measures.

Note that the correlation coefficients calculated in this study are over a large domain over East Asia and western Pacific bigger than eastern China. However, averaged over the years with moderate to high correlations (>0.5), the atmospheric circulation pattern anomalies (Figure S4 in Supporting Information S1) are almost the same to those in December 2016/2017 in eastern China (Figure 2). The available $PM_{2.5}$ observational data cover 10 years. Anomalous atmospheric circulation patterns favoring severe haze may not be completely included in this study. And we normalize the $PM_{2.5}$ with the monthly SO₂ emissions to minimize the impact of reducing emissions. However, this approach may bring some uncertainties for the $PM_{2.5}$ also composed of nitrates and other secondary organic aerosols, in addition to sulphate. Future work could use the model simulation to minimize the effects of emissions reduction by fixing emissions. Nevertheless, our results suggest that severe haze in China is shifting from the local accumulation in northern China to regional transport in southern China due to atmospheric circulation changes under the historical climate change. A joint prevention and control of air pollution is required in the current environmental situation. In the future, climate change under the sustainable and intermediate development scenarios are the ideal paths, with global effort in reducing greenhouse gases, helping for the decreasing severe haze in China.

Data Availability Statement

Observed hourly PM_{2.5} concentrations for years 2009–2019 are provided by the U.S. Embassy. The monthly SO₂ emissions (2000–2017) are from the MEIC inventory (https://zenodo.org/record/5706561, last access: November 2021). The CAM5 model is available at http://www.cesm.ucar.edu/models/cesm1.2/(last access: November 2021). Our CAM5-EAST model code and results can be made available upon request. Meteorological data from 1980 to 2019 obtained by ERA5 reanalysis data (https://www.ecmwf.int/en/forecasts/datasets/reanalysis-datasets/era5, last access: June 2021). The multi-model historical and future simulations under the Coupled Model Intercomparison Project Phase 6 (CMIP6) are from https://esgf-node.llnl.gov/search/cmip6/ (last access: November 2021).

References

- Burnett, R., Chen, H., Szyszkowicz, M., Fann, N., Hubbell, B., Pope, C. A., et al. (2018). Global estimates of mortality associated with long-term exposure to outdoor fine particulate matter. *Proceedings of the National Academy of Sciences of the United States of America*, 115, 9592–9597. https://doi.org/10.1073/pnas.1803222115
- Cai, W., Li, K., Liao, H., Wang, H., & Wu, L. (2017). Weather conditions conducive to Beijing severe haze more frequent under climate change. *Nature Climate Change*, 7, 257–262. https://doi.org/10.1038/nclimate3249
- Cao, J., Shen, Z., Chow, J. C., Watson, J. G., Lee, S., Tie, X., et al. (2012). Winter and summer PM_{2.5} chemical compositions in fourteen chinese cities. *Journal of the Air and Waste Management Association*, 62, 1214–1226. https://doi.org/10.1080/10962247.2012.701193
- Cohen, A. J., Brauer, M., Burnett, R., Anderson, H. R., Frostad, J., Estep, K., et al. (2017). Estimates and 25-year trends of the global burden of disease attributable to ambient air pollution: An analysis of data from the Global Burden of Diseases Study 2015. *Lancet*, 389, 1907–1918. https://doi.org/10.1016/s0140-6736(17)30505-6
- Ding, Y., & Liu, Y. (2014). Analysis of long-term variations of fog and haze in China in recent 50 years and their relations with atmospheric humidity. Science China Earth Sciences, 57, 36–46. https://doi.org/10.1007/s11430-013-4792-1
- Fu, Y., Tai, A. P. K., & Liao, H. (2016). Impacts of historical climate and land cover changes on fine particulate matter (PM_{2.5}) air quality in East Asia between 1980 and 2010. Atmospheric Chemistry and Physics, 16, 10369–10383. https://doi.org/10.5194/acp-16-10369-2016
- Geng, G., Zhang, Q., Tong, D., Li, M., Zheng, Y., Wang, S., & He, K. (2017). Chemical composition of ambient PM_{2.50} ver China and relationship to precursor emissions during 2005 –2012. Atmospheric Chemistry and Physics, 17, 9187–9203. https://doi.org/10.5194/acp-17-9187-2017
- Kang, H., Zhu, B., Gao, J., He, Y., Wang, H., Su, J., et al. (2019). Potential impacts of cold frontal passage on air quality over the Yangtze River Delta, China. Atmospheric Chemistry and Physics, 19, 3673–3685. https://doi.org/10.5194/acp-19-3673-2019
- Krotkov, N. A., McLinden, C. A., Li, C., Lamsal, L. N., Celarier, E. A., Marchenko, S. V., et al. (2016). Aura OMI observations of regional SO₂ and NO, pollution changes from 2005 to 2015. Atmospheric Chemistry and Physics, 16, 4605–4629. https://doi.org/10.5194/acp-16-4605-2016
- Li, H., Yang, Y., Wang, H., Li, B., Wang, P., Li, J., & Liao, H. (2021). Constructing a spatiotemporally coherent long-term PM_{2.5} concentration dataset over China during 1980–2019 using a machine learning approach. Science of the Total Environment, 765, 144263. https://doi. org/10.1016/j.scitotenv.2020.144263
- Li, K., Liao, H., Cai, W., & Yang, Y. (2018). Attribution of anthropogenic influence on atmospheric patterns conducive to recent most severe haze over eastern China. *Geophysical Research Letters*, 45, 2072–2081. https://doi.org/10.1002/2017g1076570
- Li, Q., Zhang, R., & Wang, Y. (2016). Interannual variation of the wintertime fog-haze days across central and eastern China and its relation with East Asian winter monsoon. *International Journal of Climatology*, *36*, 346–354. https://doi.org/10.1002/joc.4350

Acknowledgments

This study was supported by the National Key Research and Development Program of China (Grant 2020YFA0607803 and 2019YFA0606800) and the National Natural Science Foundation of China (Grant 41975159). Hailong Wang acknowledges the support by the U.S. Department of Energy (DOE), Office of Science, Office of Biological and Environmental Research (BER). The Pacific Northwest National Laboratory (PNNL) is operated for DOE by the Battelle Memorial Institute under contract DE-AC05-76RL01830.

- Liao, T., Wang, S., Ai, J., Gui, K., Duan, B., Zhao, Q., et al. (2017). Heavy pollution episodes, transport pathways and potential sources of PM_{2.5} during the winter of 2013 in Chengdu (China). Science of the Total Environment, 584, 1056–1065. https://doi.org/10.1016/j.scitotenv.2017.01.160
 Liu, N., Zhou, S., Liu, C., & Guo, J. (2019). Synoptic circulation pattern and boundary layer structure associated with PM_{2.5} during wintertime haze pollution episodes in Shanghai. Atmospheric Research, 228, 186–195. https://doi.org/10.1016/j.atmosres.2019.06.001
- Lu, S., He, J., Gong, S., & Zhang, L. (2020). Influence of Arctic Oscillation abnormalities on spatio-temporal haze distributions in China. Atmospheric Environment, 223, 1352–2310. https://doi.org/10.1016/j.atmosenv.2020.117282
- Mai, J., Deng, T., Yu, L., Deng, X., Tan, H., Wang, S., & Liu, X. (2016). A modeling study of impact of emission control strategies on PM_{2.5} reductions in Zhongshan, China, using WRF-CMAQ. Advances in Meteorology, 2016, 5836070. https://doi.org/10.1155/2016/5836070
- Pei, L., Yan, Z., Chen, D., & Miao, S. (2020). Climate variability or anthropogenic emissions: Which caused Beijing Haze? Environmental Research Letters, 15, 034004. https://doi.org/10.1088/1748-9326/ab6f11
- Reddington, C. L., Conibear, L., Knote, C., Silver, B. J., Li, Y. J., Chan, C. K., et al. (2019). Exploring the impacts of anthropogenic emission sectors on PM_{2.5} and human health in South and East Asia. Atmospheric Chemistry and Physics, 19, 11887–11910. https://doi.org/10.5194/ acp-19-11887-2019
- Ren, L., Yang, Y., Wang, H., Zhang, R., Wang, P., & Liao, H. (2020). Source attribution of Arctic black carbon and sulfate aerosols and associated Arctic surface warming during 1980–2018. Atmospheric Chemistry and Physics, 20, 9067–9085. https://doi.org/10.5194/acp-20-9067-2020
- Shindell, D. T., Miller, R. L., Schmidt, R. A., & Pandolfo, L. (1999). Simulation of recent northern winter climate trends by greenhouse-gas forcing. *Nature*, 399, 452–455. https://doi.org/10.1038/20905
- Sun, T., Che, H., Qi, B., Wang, Y., Dong, Y., Xia, X., et al. (2018). Aerosol optical characteristics and their vertical distributions under enhanced haze pollution events: Effect of the regional transport of different aerosol types over eastern China. Atmospheric Chemistry and Physics, 18, 2949–2971. https://doi.org/10.5194/acp-18-2949-2018
- Wang, H., Chen, H., & Liu, J. (2015). Arctic Sea Ice Decline Intensified Haze Pollution in Eastern China. Atmospheric and Oceanic Science Letters, 8, 1–9. https://doi.org/10.3878/AOSL20140081
- Wang, H., Rasch, P. J., Easter, R. C., Singh, B., Zhang, R., Ma, P. L., et al. (2014). Using an explicit emission tagging method in global modeling of source-receptor relationships for black carbon in the Arctic: Variations, sources, and transport pathways. *Journal of Geophysical Research*, 119, 12888–12909. https://doi.org/10.1002/2014JD022297
- Wu, P., Ding, Y., & Liu, Y. (2017). Atmospheric circulation and dynamic mechanism for persistent haze events in the Beijing–Tianjin–Hebei region. Advances in Atmospheric Sciences, 34, 429–440. https://doi.org/10.1007/s00376-016-6158-z
- Yang, Y., Liao, H., & Lou, S. (2016). Increase in winter haze over eastern China in recent decades: Roles of variations in meteorological parameters and anthropogenic emissions. Journal of Geophysical Research: Atmospheres, 121, 13,050-13,13. https://doi.org/10.1002/2016jd025136
- Yang, Y., Lou, S., Wang, H., Wang, P., & Liao, H. (2020). Trends and source apportionment of aerosols in Europe during 1980-2018. Atmospheric Chemistry and Physics, 20, 2579–2590. https://doi.org/10.5194/acp-20-2579-2020
- Yang, Y., Ren, L., Li, H., Wang, H., Wang, P., Chen, L., et al. (2020). Fast climate responses to aerosol emission reductions during the COVID-19 pandemic. *Geophysical Research Letters*, 47, e2020GL089788. https://doi.org/10.1029/2020gl089788
- Yang, Y., Smith, S. J., Wang, H., Mills, C. M., & Rasch, P. J. (2019). Variability, timescales, and nonlinearity in climate responses to black carbon emissions. Atmospheric Chemistry and Physics, 19, 2405–2420. https://doi.org/10.5194/acp-19-2405-2019
- Yang, Y., Wang, H., Smith, S. J., Easter, R., Ma, P.-L., Qian, Y., et al. (2017). Global source attribution of sulfate concentration and direct and indirect radiative forcing. Atmospheric Chemistry and Physics, 17, 8903–8922. https://doi.org/10.5194/acp-17-8903-2017
- Yang, Y., Wang, H., Smith, S. J., Ma, P.-L., & Rasch, P. J. (2017). Source attribution of black carbon and its direct radiative forcing in China. Atmospheric Chemistry and Physics, 17, 4319–4336. https://doi.org/10.5194/acp-17-4319-2017
- Yang, Y., Wang, H., Smith, S. J., Zhang, R., Lou, S., Qian, Y., et al. (2018). Recent intensification of winter haze in China linked to foreign emissions and meteorology. *Scientific Reports*, 8, 2107. https://doi.org/10.1038/s41598-018-20437-7
- Yang, Y., Wang, H., Smith, S. J., Zhang, R., Lou, S., Yu, H., et al. (2018). Source apportionments of aerosols and their direct radiative forcing and long-term trends over continental United States. *Earth's Future*, 6, 793–808. https://doi.org/10.1029/2018EF000859
- Zhang, Q., Jiang, X., Tong, D., Davis, S. J., Zhao, H., Geng, G., et al. (2017). Transboundary health impacts of transported global air pollution and international trade. *Nature*, 543, 705–709. https://doi.org/10.1038/nature21712
- Zhang, R., Li, Q., & Zhang, R. (2014). Meteorological conditions for the persistent severe fog and haze event over eastern China in January 2013. *Science China Earth Sciences*, 57, 26–35. https://doi.org/10.1007/s11430-013-4774-3
- Zhang, Y., Ding, A., Mao, H., Nie, W., Zhou, D., Liu, L., et al. (2016). Impact of synoptic weather patterns and inter-decadal climate variability on air quality in the North China Plain during 1980–2013. Atmospheric Environment, 124, 119–128. https://doi.org/10.1016/j. atmosenv.2015.05.063
- Zhao, S., Li, J., & Sun, C. (2016). Decadal variability in the occurrence of wintertime haze in central eastern China tied to the Pacific Decadal Oscillation. *Scientific Reports*, *6*, 27424. https://doi.org/10.1038/srep27424
- Zheng, B., Tong, D., Li, M., Liu, F., Hong, C., Geng, G., et al. (2018). Trends in China's anthropogenic emissions since 2010 as the consequence of clean air actions. Atmospheric Chemistry and Physics, 18, 14095–14111. https://doi.org/10.5194/acp-18-14095-2018
- Zhong, W., Yin, Z., & Wang, H. (2019). The relationship between anticyclonic anomalies in northeastern Asia and severe haze in the Beijing– Tianjin–Hebei region. Atmospheric Chemistry and Physics, 19, 5941–5957. https://doi.org/10.5194/acp-19-5941-2019
- Zou, Y., Wang, Y., Zhang, Y., & Koo, J. H. (2017). Arctic sea ice, Eurasia snow, and extreme winterhaze in China. Science Advances, 3, e1602751. https://doi.org/10.1126/sciady.1602751